\begin{table}%t1
    \caption {\label{tab:datasum}Summary of the subsets of VHE data used in this
  paper (\mjd~=~MJD-53~900). 
  }
%\centerline
 {
\begin{tabular}{lccccccl}
\hline
\hline
\\
Label$^a$     
&   \mjd start &  \mjd end  &   Duration & En. thres.  & Fig.$^b$ 
& Sect.$^b$ & Notes \\
         &               &            &    [hrs]   &    [GeV]       &         &    &  \\
\hline
\texttt{T200}  &  45.934618  &  46.107844  & 4.16  & 200  &\ref{fig:lcs2min}& Sect.~3.1 & \\
\texttt{T300}  &  45.913536  &  46.129121  & 5.17  & 300  &\ref{fig:lcs2min}& Sect.~3.1 &\\
\texttt{T300-X}&  45.923252  &  46.129121  & 4.94  & 300  &\ref{fig:xg300}  & Sect.~3.3 & {\texttt{T300} with X-ray coverage}\\
\texttt{T400}  &  45.892483  &  46.150125  & 6.18  & 400  &\ref{fig:lcs2min}& Sect.~3.1& \\
\texttt{T500}  &  45.871450  & 46.171258   & 7.20  & 500  &\ref{fig:lcs2min}& Sect.~3.1& \\
\texttt{T700}  &  45.850393  &  46.171258  & 7.70  & 700  &\ref{fig:lcs2min}& Sect.~3.1& \\
            &             &             &       &      &            &  & \\
\texttt{T400-Peak}  &  45.896643  & 45.920474	& 0.57  & 400  &\ref{fig:fluxindex500}& Sect.~4.1& {Spectrum at the peak of VHE flare} \\
\texttt{T300-High}  &  45.913530  &   45.970312 & 1.36  & 300  &\ref{fig:fluxindex300}& Sect.~4.1& {Average high-state spectrum inside \texttt{T300}.} \\
\texttt{T300-Low}   &  46.013252  &   46.129166 & 2.78  & 300  &\ref{fig:fluxindex300}& Sect.~4.1& {Average low-state spectrum inside \texttt{T300}.}\\
            &             &             &       &      &            &  & \\
\texttt{T300-Xmax} &  45.922130  &  45.944699  & 0.54  & 300  &\ref{fluxgo}& Sect.~4.2& 
{Highest simultaneous X-ray/VHE state.}  \\
\texttt{T400-Xmin} &  46.100000  &  46.150000  & 1.20  & 400  &\ref{fluxgo}& Sect.~4.2 & {Lowest simultaneous X-ray/VHE state.}\\
\texttt{T300-RXTE} &  46.016846  &  46.033328  &       &      &  	  &  & \\
                   &  46.084624  &  46.098698  & 0.73  & 300  &\ref{fluxgo}& Sect.~4.1 &
                   {Sum of the 2 intervals with RXTE coverage.} \\
\\
\hline
%\hline
\end{tabular}}
\par
\smallskip
$^a$ {The T-label corresponds to the VHE energy threshold of that dataset.}
$^b$ {Figure where the 
time intervals are indicated and the section where they are first introduced.}
\end{table}